Author's response to reviews

Title: Imputation by the mean score should be avoided when validating a Patient Reported Outcomes questionnaire by a Rasch model in presence of informative missing data

Authors:

Jean-Benoit Hardouin (jean-benoit.hardouin@univ-nantes.fr)
Ronan Conroy (rconroy@rcsi.ie)
Véronique Sébille (veronique.sebille@univ-nantes.fr)

Version: 4 Date: 13 July 2011

Author's response to reviews:

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format:
Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

OK

Equations: Please note that the PDF version of articles in this journal is laid out in a two-column format. We therefore request that you break long equations so that they can be rendered legibly within a 85mm column width.

OK, there is no long equations

Structure: this manuscript does not have the correct structure. Please reformat it to use these headings:
Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Authors’ information (optional); Acknowledgements (optional); References

We don’t understand this remark because the structure of the paper was already this one (with subsections in several sections as Methods, Results or Discussion). Only the former Discussion has been cut between a discussion and a conclusion

Authors’ contributions: Please indicate that all authors read and approved the final manuscript.

OK

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

The highlighting have been deleted

Tables- please remove the visible vertical lines from your tables

We prefer you do this modification, because we find the tables are very hard to read without vertical lines

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

The manuscript has been checked.